UTILITY OF THE WORLD HEALTH ORGANIZATION DISABILITY ASSESSMENT SCHEDULE AND THE WORLD HEALTH ORGANIZATION MINIMAL GENERIC SET OF DOMAINS OF FUNCTIONING AND HEALTH IN SPINAL CORD INJURY

Sinikka TARVONEN-SCHRÖDER, MD, PhD¹, Anne KALJONEN, BSc¹ and Katri LAIMI, MD, PhD¹ From the ¹Department of Rehabilitation and Brain Trauma, Division of Clinical Neurosciences and ³Department of Physical and Rehabilitation Medicine, Turku University Hospital and University of Turku, and ²Department of Biostatistics, University of Turku, Turku, Finland

Objective: To compare easy-to-use International Classification of Functioning, Disability and Health (ICF)-based measures of functioning with the level and severity of spinal cord injury.

Methods: Cross-sectional study. Patients (n = 142)and their significant others completed the 12-item World Health Organization Disability Assessment Schedule (WHODAS 2.0) questionnaire. A physician at the university hospital outpatient clinic assessed functioning with the 7-item World Health Organization (WHO) minimal generic set.

Results: The patient and proxy WHODAS sum score was rated severe with decreasing severity in groups with complete and partial tetraplegia and paraplegia, respectively. Working ability was rated most severely impaired in the tetraplegic groups. Between-group differences were also found in mobility, household tasks, and self-care. Mobility was found to be associated with lesion severity; life activities, participation and friendships with lesion level; and self-care and WHODAS sum score with both lesion severity and level. Depending on the level and severity of spinal cord injury, a moderate to strong correlation was found between the sum scores of the 2 tools, and mostly very strong correlations between patient and proxy assessments of functioning.

Conclusion: Both generic ICF-based tools, despite their briefness, seemed to be useful as they were able to differentiate various levels and severities of spinal cord injury. We recommend using the 12-item WHODAS 2.0 when planning individualized services for patients with spinal cord injury.

Key words: disability; ICF; severity; spinal cord injury; WHO minimal generic set of domains of functioning and health; WHODAS.

Accepted Oct 3, 2018; Epub ahead of print Nov 8, 2018

J Rehabil Med 2019; 51: 40-46

Correspondence address: Sinikka Tarvonen-Schröder, Division of Clinical Neurosciences, Turku University Hospital, PO Box 52, FIN-20521 Turku, Finland. E-mail: sinikka.tarvonen-schroder@tyks.fi

Spinal cord injury (SCI) is a major cause of longterm disability (1–9), with increasing prevalence rates unless measures are taken to control incidence (2). It has been estimated that the prevalence of nontraumatic SCI may overtake traumatic SCI in the next

LAY ABSTRACT

Two easy-to-use validated International Classification of Functioning, Disability and Health (ICF)-based tools, the 12-item patient and proxy World Health Organization Disability Assessment Schedule (WHODAS 2.0) and the 7-item World Health Organization minimal generic set used by a physician, were found to be reliable and useful in assessing functioning in patients with spinal cord injury. These tools can, with little effort, be used to detect activity limitations and participation restrictions, and differentiate various severities and levels of spinal cord injury. As the minimal generic set is brief and limited, we recommend using the 12-item WHODAS 2.0 when planning individualized services for patients with spinal cord injury.

decades (6). In the light of previous studies, patients with non-traumatic SCI seem to reach the same functional outcomes as those with traumatic SCI (1, 4). With improved acute care and greater life expectancies after SCI, the focus of management has shifted to interventions aiming to improve functioning, quality of life and involvement in life situations. The ultimate aim of rehabilitation is for patients to reach optimal participation despite their disabilities.

Different instruments have been used to evaluate the severity of SCI, patient's needs and treatment effects (10–13). Some of these outcome measures do not address functional relevance (11). Priorities in developing appropriate functioning instruments for patients with SCI include understanding the dimensions of participation, and the importance of both objective and subjective measures (10). Only a few studies, however, have tested participation instruments, such as WHODAS 2.0, in individuals with SCI, and there is still a lack of studies comparing self-rated functioning with more objectively defined parameters (14–19). Other difficulties in measuring functioning lie in a redundancy of instruments for different health conditions and tools that are too time-consuming for clinical settings. Among ICF-based tools developed by the WHO (20), the 2 shortest validated generic assessment scales are the 7-item minimal generic set of domains of functioning and health (21) and the 12item World Health Organization Disability Assessment Schedule (WHODAS 2.0) (22). These 7- and 12-item

scales evaluate different items in the "activities and participation" section of the ICF. While participation is a crucial outcome of SCI rehabilitation (10, 23–25), brief validated measures to evaluate participation are needed.

As far as we know, there are no studies comparing these 2 tools, the WHODAS 2.0 and the WHO minimal generic set, in assessing functioning in patients with SCI of different severities and injury levels. The objective of this study was to compare the utility of these 2 brief scales in assessing functioning in SCI.

PATIENTS AND METHODS

This observational cross-sectional study was conducted between November 2015 and November 2017. The 12-item WHODAS 2.0 and personal background information (age, sex, accommodation, marital status, educational level, and working status) questionnaires, along with informed consent and information regarding the study were posted to 240 consecutively referred patients with SCI (diagnosis according to the ICD-10 criteria) and their significant others 2 weeks before their visit at a specialist outpatient clinic of a university hospital. The outpatient clinic both evaluates patients with recent SCI and takes care of patients with past SCI with ongoing problems and rehabilitative needs. A total of 160 patients responded, of whom 18 were excluded because they did not fulfil the inclusion criteria (age under 18 years at the time of the SCI, a current major medical or psychotic condition or another neurological diagnosis, and inability to understand, co-operate, or complete the questionnaire or complete recovery without residual symptoms). Of the 142 participants, 85 had also responses from their significant others; of whom 61 (72.6%) were spouses, 3 (3.6%) parents, 6 (7.1%) children, 3 (3.6%) siblings, 1 (1.2%) other relative, 3 (3.6%) close friends, and 7 (8.3%) trained caregivers.

Clinical information (date of diagnosis and comorbidities) was gathered from the hospital records, and the total number of comorbidities was counted (26, 27). The level and severity (grade) of SCI was classified according to the American Spinal Cord Injury Association (ASIA) (28). In this article, the term tetraplegia is used to describe the change in motor and/or sensory function when the injury is at the level of the cervical segments. The term paraplegia is used when the lesion is more caudal. A specialist in physical and rehabilitation medicine at the outpatient clinic completed the 7-item WHO minimal generic set aiming at assessing the level of functioning in a simple and validated way (21).

The 12-item patient and proxy WHODAS 2.0 includes 12 items assessing 6 disability domains during the previous 30 days: cognition (learning and concentration), mobility (standing and walking), self-care (washing and dressing oneself), getting along (dealing with strangers and maintaining friendships), life activities (doing housework and working ability), and participation (emotional functions and engaging in community). Each of these 12 items is rated according to a 5-point Likert-type scale, which grades the difficulty experienced by a participant in performing a given activity. Each of the 12 items is scored from 0 to 4, where 0 means no (0–4%), 1 means mild (5–24%), 2 means moderate (25–49%), 3 means severe (50–95%), and 4 means extreme or complete (96–100%) difficulty in this specific activity. The total score of WHODAS is the sum of all

these 12 sub-scores, ranging from 0 to 48, with lower scores indicating better functioning. Total scores of 1–4 indicate mild disability, 5–9 moderate disability, and 10–48 severe disability (14, 15, 22, 29, 30).

The WHO minimal generic set consists of 7 ICF domains: energy and drive functions, emotional functions, sensation of pain, carrying out daily routine, walking, moving around, and remunerative employment. Generic means that this assessment scale is applicable to all people despite their health conditions. Minimal means that the scale consists of the least number of domains of functioning that can be used to explain significant differences between people with health issues. The scoring system is similar to WHODAS, the sum score ranging from 0 to 28, with lower scores indicating better functioning (21).

Statistical analysis

The comparison between the 4 patient groups according to severity and level (tetraplegia AIS A-B and C-D and paraplegia AIS A-B and C-D) was carried out within categorical variables using χ^2 test, or, in the case of small cell frequencies, Fisher's exact test. In numerical variables the comparisons between the 4 patient groups was carried out either by 1-way analysis of variance (ANOVA), or by the non-parametric Kruskal-Wallis test if the distribution of an outcome variable was too skewed. The Spearman's correlation coefficient was used to test the correlation between variables. Patient and proxy WHODAS responses were analysed pairwise; the patients with only 1 value were excluded. The correlations of 0-0.30 were considered weak, 0.31-0.50 moderate, 0.51-0.70 strong, and greater than 0.70 very strong. No adjustment for multiple comparisons was done, since in this exploratory study the search for patterns is more important than formal statistical significance. In order to reach enough statistical power we set a minimum target number of patients to 140. Statistical analyses were performed using SAS 9.4 for Windows. p-values below 0.05 (2-tailed) were considered statistically significant.

RESULTS

Demographic and clinical background data for all 142 participants are shown in Table I. To compare background data in different levels and severity groups of SCI, participants were divided into 4 groups: (*i*) complete tetraplegia AIS A–B (n=13), (*ii*) partial tetraplegia AIS C–D (n=58), (*iii*) complete paraplegia AIS A–B (n=23), and (*iv*) partial paraplegia AIS C–D (n=48).

When disability was rated by a physician (Table II), disabilities were severe in all patient groups, and patients with a complete tetraplegia were rated the most disabled. In all groups, energy and emotional functions were rated mildly impaired, pain moderately, daily activities at least moderately, and employment severely impaired. The total disability (sum score of the generic set) and 4 of the 7 separate functions were more impaired in those with a complete motor plegia (AIS A–B, n=36) than in those with a partial plegia (AIS C–D, n=106). When those with a cervical lesion

Journal of Rehabilitation Medicine

42 S. Tarvonen-Schröder et al.

Variables	Total (<i>n</i> = 142)	Tetraplegia AIS A-B (n=13)	Tetraplegia AIS C-D (n = 58)	Paraplegia AIS A-B (n=23)	Paraplegia AIS C-D (n=48)	<i>p</i> -value
Age, years, mean (SD)	56.7 (16.9)	52.3 (16.5)	61.2 (15.8)	53.2 (18.6)	60.0 (16.6)	ns
Years since diagnosis, mean (SD)	8.8 (9.8)	12.1 (10.8)	6.4 (9.3)	10.6 (10.5)	6.2 (8.4)	ns
Number of comorbidities, mean (SD)	1.1 (1.0)	0.8 (0.8)	1.1(1.1)	1.1(1.1)	1.4 (1.2)	ns
Charlson comorbidity index, mean (SD)	0.5 (0.7)	0.5 (0.7)	0.3 (0.7)	0.6 (0.9)	0.6 (0.8)	ns
Education in years, mean (SD)	11.5 (2.6)	11.5 (2.4)	12.4 (4.0)	10.6 (2.8)	11.4 (3.0)	ns
Sex, male, <i>n</i> (%)	75 (52.8)	10 (76.9)	28 (48.3)	13 (56.5)	24 (50.0)	ns
Still working, n (%)	19 (13.4)	1 (7.7)	8 (13.8)	2 (8.7)	8 (16.7)	ns
Cohabiting, n (%)	82 (57.8)	5 (38.5)	31 (53.4)	14 (60.9)	32 (66.7)	ns
Institutionalized, n (%)	8 (5.6)	4 (30.8)	3 (5.2)	1 (4.4)	0 (0.0)	
Aetiology						
Traumatic	57 (40.1)	10 (76.9)	17 (29.3)	16 (69.6)	14 (29.2)	0.0001
Spinal stenosis/ myelopathy/tumour	74 (52.1)/8 (5.7)/3 (2.1)					
AIS grade A/B/C/D	27 (19.0)/9 (6.3)/17 (12.0)/89 (62	2.7)				
Segment C1-4/C5-7/Th1-11	6 (4.2) ^a /7 (4.9)/107 (75.4)					
Th12-S2/S3-5	18 (12.7)/4 (2.8)					

^aThree with ventilatory aid.

AIS: ASIA (American Spinal Cord Injury Association) Impairment Scale; ns: not significant; SD: standard deviation.

Table II. Physician-rated generic set scores by severity and level of spinal cord injury

Variable	Tetraplegia AIS A-B (n=13) Mean (SD)	Tetraplegia AIS C-D (<i>n</i> = 58) Mean (SD)	Paraplegia AIS A-B (<i>n</i> = 23) Mean (SD)	Paraplegia AIS C-D (n=48) Mean (SD)	<i>p</i> -value
Generic set sum score (0–28)	25.2 (6.1)	15.6 (4.0)	18.4 (2.9)	15.0 (4.0)	< 0.001
Energy and drive functions (0-4)	0.9 (0.8)	1.2 (0.8)	1.1 (0.8)	1.0 (0.8)	ns
Emotional functions	1.1 (1.0)	1.2 (0.8)	1.2 (0.8)	1.1 (0.9)	ns
Sensation of pain	1.8 (1.2)	1.9 (1.0)	1.8 (1.1)	1.9 (0.9)	ns
Daily activities	2.9 (1.2)	1.9 (1.0)	2.4 (0.7)	1.6 (0.8)	0.0001
Walking	4.0 (0.0)	2.3 (1.0)	4.0 (0.2)	2.5 (0.9)	< 0.0001
Moving around	4.0 (0.0)	3.4 (0.9)	4.0 (0.0)	3.5 (0.6)	< 0.001
Remunerative employment	3.8 (0.6)	3.6 (0.8)	3.8 (0.5)	3.5 (1.0)	ns

Score of each item 0-4, 0 = no difficulty, 4 = total inability.

AIS: ASIA (American Spinal Cord Injury Association) Impairment Scale.

(tetraplegia, n=71) were compared with those with a more caudal lesion (paraplegia, n=71), no betweengroup differences were found.

When functioning was rated by patients and their significant others (WHODAS 2.0, Table III), the total disability score showed severe impairment in all patient groups. Difficulties reported in cognition, participation, and relationships remained mild, with the exception of moderate impairment in community life in those with a partial tetraplegia. All patient groups reported severe difficulties in standing, walking, and working ability, the restrictions being most severe in those with tetraplegia. When comparing tetraplegic and paraplegic groups, similar between-group differences were also

Table III. Functioning assessed	I with patient and prox	y WHODAS 2.0 by severi	ty and level of injury
---------------------------------	-------------------------	------------------------	------------------------

	Rated by patients				Rated by significant others					
Variables	Tetraplegia AIS A-B (n=13) Mean (SD)	Tetraplegia AIS C-D (n = 58) Mean (SD)	Paraplegia AIS A-B (n = 23) Mean (SD)	Paraplegia AIS C-D (n = 48) Mean (SD)	<i>p</i> -value	Tetraplegia AIS A-B (n = 10) Mean (SD)	Tetraplegia AIS C-D (n = 29) Mean (SD)	Paraplegia AIS A-B (n = 17) Mean (SD)	Paraplegia AIS C-D (n = 29) Mean (SD)	<i>p</i> -value
Total score (0-48)	25.2 (6.1)	21.0 (10,2)	18.8 (6.1)	16.3 (8.8)	< 0.01	25.9 (8.6)	21.6 (9.7)	19.7 (6.5)	17.8 (9.5)	ns
Standing (0-4)	3.9 (0.3)	2.9 (1.4)	4.0 (0.0)	3.0 (1.3)	0.0001	3.9 (0.3)	2.9 (1.2)	4.0 (0.0)	2.9 (1.4)	< 0.01
Household tasks	3.2 (1.3)	2.6 (1.3)	1.9 (1.4)	2.1 (1.3)	< 0.01	2.8 (1.6)	2.6 (1.4)	2.1 (1.5)	2.3 (1.4)	ns
Learning	0.2 (0.4)	0.3 (0.5)	0.2 (0.4)	0.2 (0.4)	ns	0.4 (0.7)	0.2 (0.5)	0.3 (0.6)	0.3 (0.7)	ns
Community life	0.9 (1.4)	1.8 (1.4)	1.1 (1.0)	1.4 (1.1)	< 0.05	1.0 (1.3)	2.0 (1.1)	0.9 (0.9)	1.7 (1.3)	< 0.05
Emotional functions	1.6 (1.4)	1.8 (1.1)	1.2 (1.0)	1.3 (1.0)	ns	1.7 (1.6)	2.0 (1.1)	1.2 (0.8)	1.5 (1.1)	ns
Concentrating	0.3 (0.8)	0.7 (1.2)	0.5 (0.9)	0.5 (0.8)	ns	0.6 (1.3)	0.7 (0.9)	0.5 (0.8)	0.6 (0.9)	ns
Walking	4.0 (0.0)	2.9 (1.4)	4.0 (0.0)	2.8 (1.5)	0.0001	4.0 (0.0)	2.8 (1.3)	4.0 (0.0)	3.0 (1.3)	< 0.001
Washing oneself	3.5 (0.9)	1.9 (1.5)	1.4 (1.2)	1.3 (1.4)	< 0.0001	3.6 (0.7)	1.9 (1.4)	1.9 (1.4)	1.4 (1.6)	< 0.01
Dressing oneself	3.2 (1.1)	1.8 (1.4)	1.4 (1.2)	1.3 (1.4)	< 0.0001	3.2 (1.1)	1.9 (1.4)	1.6 (1.4)	1.3 (1.4)	< 0.01
Dealing with strangers	0.8 (1.0)	0.5 (0.9)	0.3 (0.9)	0.3 (0.7)	ns	0.8 (1.2)	0.6 (0.9)	0.2 (0.6)	0.4 (0.7)	ns
Maintaining friendships	0.7 (1.0)	1.1 (1.3)	0.6 (0.9)	0.4 (0.6)	< 0.01	1.1 (1.2)	1.2 (1.4)	0.8 (1.1)	0.4 (0.6)	ns
Working ability	2.9 (1.7)	2.8 (1.5)	2.2 (1.6)	2.1 (1.5)	ns	2.8 (1.7)	2.9 (1.4)	2.3 (1.6)	1.9 (1.6)	ns

Score of each item 0-4, 0 = no difficulty, 4 = total inability.

AIS: ASIA (American Spinal Cord Injury Association) Impairment Scale; ns: not significant.

JRM Journal of Rehabilitation Medicine JRM

Variable	AIS grade D $(n = 89)$ vs grades A-C $(n = 53)$	Lesion level tetraplegia (n = 71) vs paraplegia (n = 71)	Any walking ability (n = 100) vs seated (n = 42)	Current employment (n=19)	Generic set sun $(n = 142)$
Cognition	0.15	-0.04	0.21*	0.09	0.08
Mobility	-0.52****	0.09	-0.52****	0.34****	0.56****
Self-care	-0.24**	-0.30***	-0.25**	0.26**	0.39****
Getting along	-0.04	-0.19*	-0.01	0.11	0.26**
Life activities	-0.04	-0.27**	-0.06	0.24**	0.40****
Participation	0.07	-0.18*	0.19*	0.13	0.27***
WHODAS sum score	-0.16*	-0.24**	-0.15	0.27***	0.49****
Generic set sum score	-0.33****	0.02	-0.41****	0.33****	1.00

Table IV. Spearman Correlations of patient-rated World Health Organization Disability Assessment Schedule (WHODAS) domains with spinal cord injury severity (AIS grade), level, walking ability, working status, and physician-rated functioning

*p<0.05, **p<0.01, ***p<0.001, ****p<0.001.

AIS: ASIA (American Spinal Cord Injury Association) Impairment Scale.

found in household activities and self-care. When the ratings of those with a complete lesion (AIS A-B) were compared with those with a partial lesion (AIS C–D), a complete lesion was associated with more difficulties in getting around and self-care, and a partial lesion with participation difficulties.

The Spearman correlations of the 6 patient-rated WHODAS domains with SCI severity (AIS grade), lesion level, walking ability, current working status, and physician-rated functioning (generic set sum score) are shown in Table IV. The correlation between the patient-rated WHODAS sum score and the physicianrated generic set sum score was moderate in all other patient groups, except for a strong correlation in the group with AIS D tetraplegia. The generic set sum score was correlated with all other WHODAS domains (0.26-0.56) except cognition.

The correlation coefficient between patient and proxy WHODAS sum scores was very strong (0.89) with strong to very strong correlations in single items except for the item "dealing with strangers" with only a moderate correlation. The significant others rated both the total disability (sum score) and the items learning, emotions and dressing as slightly more impaired than did the patients themselves. Cronbach's alpha value for reliability of the WHODAS patient sum was 0.86, the WHODAS significant other 0.86 and the minimal generic set sum 0.75.

DISCUSSION

These results support the utility of both generic measures WHODAS 2.0 and the WHO minimal generic set in assessing functioning of patients with SCI. The degree of perceived impairment (WHODAS sum score) was found to increase from incomplete paraplegia through complete paraplegia and partial tetraplegia to complete tetraplegia, respectively.

In our study population, participants typically had hardly any problems in cognition, emotions or relationships. The most severe impairments were found in getting around and life activities, followed by selfcare, these results being largely in line with previous studies in SCI (15-17, 31, 32). Participation, on the other hand, was rated less severely impaired among our participants, the restriction being only mild in all other groups except those with a partial tetraplegia. Obviously, the ratings do not strictly measure the level of functional capacity, but evaluate the level of actual performance in real-life situations and also reflect environmental factors, e.g. the amount of assistance and aids, support and services the patients receive in the community, as in previous studies (13). Unexpectedly, those with an incomplete SCI lesion were found to experience more difficulties in community life than those with a complete lesion. Our hypothesis is that, if people with a complete lesion and more severe symptoms get daily help from trained caregivers, those with a milder injury with no such aid may experience more restricted participation than those with a more severe or complete injury. The groups with a partial lesion were also 7-9 years older than the other 2 groups with a complete lesion, which could have some influence on the results even if the age difference was not statistically significant.

These results show that participants with a cervical lesion (partial or complete tetraplegia) experience more impairments in self-care and life activities, including the ability to work, than those with a lower lesion, which is in agreement with previous studies (18, 33–35). It is natural that, when those with a complete motor lesion were compared with those with an incomplete lesion, a complete lesion was associated with more difficulties in getting around and self-care. In our study, inability to walk was correlated with other impairments (getting around, self-care, participation and cognition) as in previous studies, which found wheelchair-dependency (36, 37) and mobility restrictions (17) to increase difficulties in other domains of participation (17, 37) and to decrease quality of life (36). Also, current employment was found to be associated with other domains (mobility, self-care, and life

activities) in our study population. These findings are, in many aspects, in agreement with the only previous WHODAS study evaluating these associations between objective and subjective parameters in SCI (17).

The correlation between the assessments made by a physician (the WHO minimal generic set) and by the patients themselves (WHODAS) was moderate to strong. This study is in line with previous studies showing this kind of positive correlation between WHODAS 2.0 and other measures of activity limitations (30), in our study the WHO minimal generic set. As the correlations between the patient and proxy WHODAS scores on the whole and in single items were mostly very strong, it seems possible that, when assessing functioning in patients with SCI who do not have cognitive problems or concomitant neurological diseases, as in our study, the patient is able to make assessments reliably without reflecting his or her opinion with that of his or her significant others.

Although the physician who rated functioning using the generic set score was not blinded to background information, their ratings seemed reasonable, and might be even more reliable due to their good insight into the clinical history of the participants. Physicianrated functioning was correlated more clearly with lesion severity, walking ability and current employment status than when rated by the patients themselves. A physician rated the overall functioning severely impaired in tetraplegia and more moderately in the other 3 groups, showing between-group differences also in mobility and daily activities. The fact that working ability was rated more impaired by a physician than by the patients is interesting. Few of the patients were actually employed. Patients may have a more subjective view of employment possibilities and the functioning ability they still possess with diverse voluntary work possibilities, as a physician may think more of the real requirements in the open labour market reflected with the activity restrictions and participation limitations the patients may have.

In this study, both WHODAS and the WHO minimal generic set were found to differentiate varying severities and levels of SCI from each other. Previously, WHODAS has been shown to be useful in assessing disability in several health conditions, both in psychiatric and somatic conditions. While the burden of more time-consuming functioning tools is challenging for clinicians and patients, these simple tools appeared to be practical and useful in our clinical setting. In the wide field of rehabilitation of patients with SCI, developing cost-effective client-centred rehabilitative services is important. In this development, it is essential to assess perceived functioning, patients' needs and participation restrictions using reliable tools. Although previous recommendations regarding evaluating functioning have suggested using the 7-item WHO minimal generic set together with other wider ICF-based tools, based on our results we would also recommend using the 12-item WHODAS 2.0 alone. It is, however, possible that the 12-item WHODAS is not adequate in finding between-group differences in mobility of patients with SCI, as the description of the mobility items in WHODAS includes "walking a long distance" and "standing a long time", which are not usually possible for any patients with SCI (38). In contrast, the 7-item minimal generic set does not define the time or the distance of these activities thus allowing better clinical adjustment of moving difficulties than WHODAS. Even if WHODAS has many advantages, it does not substitute for other generic measures of both capacity and performance (capability without and with assistance) in real-life situations. In an outpatient clinic, as in our study, keeping the number of separate items minimal probably ensures the compliance of patients to complete the assessment scale. The 7-item WHO minimal generic set may, however, be too brief if used as the only functioning measure. When planning individual rehabilitative services for patients with SCI and their significant others, we primarily recommend using the 12-item WHODAS 2.0 as a screening tool.

Limitation

One limitation of the study is a cross-sectional design preventing confirming causal relationships of disability. Even if we included all consecutive 142 patients, the study population of a tertiary outpatient clinic is always selected. Even if the severity and level of the SCI was evaluated retrospectively, we had access to electronic patient records from the beginning of the medical history. As 2 different generic functioning scales (WHODAS and the WHO minimal generic set) were used, straight comparisons were not possible for all sub-items. Both measures are, however, ICF-based and have many identical items. Even if WHODAS and the WHO minimal generic set seemed to be adequate in the chronic phase as in our study, these results are not directly generalizable to the acute or subacute phase (39). Patients and their significant others were not blinded to each others' evaluations; however, in the vast majority their responses differed from each other.

Conclusion

Both ICF-based tools were easy to use in assessing functioning in SCI. Of these 2 tools, we recommend using the 12-item patient-completed WHODAS 2.0, as this can, with little effort, be used to detect activity limitations and participation restrictions in SCI, and to differentiate between various severities and levels of SCI, when planning rehabilitative services.

ACKNOWLEDGEMENTS

The study was approved by the ethics committee of the University and University Hospital (19.5.2015, 73/2015). The ethical standards of the World Medical Association Helsinki Declaration of 1975, as revised in 1983, were followed.

The authors have no conflicts of interest to declare.

REFERENCES

- 1. McKinley WO, Seel RT, Hardman JT. Nontraumatic spinal cord injury: incidence, epidemiology, and functional out-come. Arch Phys Med Rehabil 1999; 80: 619–623.
- 2. O'Connor PJ. Prevalence of spinal cord injury in Australia. Spinal Cord 2005; 43: 42–46.
- Noonan VK, Fingas M, Farry A, Baxter D, Singh A, Fehlings MG, et al. Incidence and prevalence of spinal cord injury in Canada: a national perspective. Neuroepidemiology 2012; 38: 219–226.
- van den Berg ME, Castellote JM, Mahillo-Fernandez I, de Pedro-Cuesta J. Incidence of nontraumatic spinal cord injury: a Spanish cohort study (1972–2008). Arch Phys Med Rehabil 2012; 93: 325–331.
- Adriaansen JJ, Post MW, de Groot S, van Asbeck FW, Stolwijk-Swuste JM, Tepper M, et al. Secondary health conditions in persons with spinal cord injury: a longitudinal study from one to five years post-discharge. J Rehabil Med 2013; 45: 1016–1022.
- New PW, Farry A, Baxter D, Noonan VK. Prevalence of nontraumatic spinal cord injury in Victoria, Australia. Spinal Cord 2013; 51: 99–102.
- 7. New PW, Cripps RA, Bonne Lee B. Global maps of nontraumatic spinal cord injury epidemiology: towards a living data repository. Spinal Cord 2014; 52: 97–109.
- Singh A, Tetreault L, Kalsi-Ryan S, Nouri A, Fehlings MG. Global prevalence and incidence of traumatic spinal cord injury. Clinical Epidemiology 2014; 6: 309–331.
- Adriaansen JJ, Ruijs LE, van Koppenhagen CF, van Asbeck FW, Snoek GJ, van Kuppevelt D, et al. Secondary health conditions and quality of life in persons living with spinal cord injury for at least ten years. J Rehabil Med 2016; 48: 853–860.
- Magasi SR, Heinemann AW, Whiteneck GG. Participation following traumatic spinal cord injury: an evidencebased review for research. J Spinal Cord Med 2008; 31: 145–156.
- Alexander MS, Anderson K, Biering-Sorensen F, Blight AR, Brannon R, Bryce T, et al. Outcome measures in spinal cord injury: recent assessments and recommendations for future directions. Spinal Cord 2009; 47: 582–591.
- Furlan JC, Noonan V, Singh A, Fehlings MG. Assessment of impairment in patients with acute traumatic spinal cord injury: a systematic review of the literature. J Neurotrauma 2011; 28: 1445–1477.
- Barclay L, McDonald R, Lentin P. Social and community participation following spinal cord injury: a critical review. Int J Rehabil Res 2015; 38: 1–19.
- Noonan V, Kopec J, Noreau L, Singer J, Mâsse L, Zhang H, et al. Measuring participation among persons with spinal cord injury: comparison of three instruments. Topics Spinal Cord Inj Rehabil 2010; 15: 49–62.
- Wolf de AC, Tate RL, Lannin NA, Middleton J, Lane-Brown A, Cameron ID. The World Health Organization Disability Assessment Scale, WHODAS II: reliability and validity in

the measurement of activity and participation in a spinal cord injury population. J Rehabil Med 2012; 44: 747–755.

- van der Zee CH, Post MW, Brinkhof MW, Wagenaar RC. Comparison of the Utrecht Scale for Evaluation of Rehabilitation-Participation with the ICF Measure of Participation and Activities Screener and the WHO Disability Assessment Schedule II in persons with spinal cord injury. Arch Phys Med Rehabil 2014; 95: 87–93.
- 17. Steinerte V, Vetra A. The World Health Organisation Disability Assessment Scale (WHODAS II): Links between self-rated health and objectively defined and clinical parameters in the population of spinal cord injury. HSH Web of Conferences 2016; 30, 00042. Int. Conf. SOCIETY. HEALTH. WELFARE. 2014. Available from: https://doaj. org/article/e0ef391fe0e247a09782573bebb80e7d.
- Noreau L, Fougeyrollas P. Long-term consequences of spinal cord injury on social participation: the occurrence of handicap situations. Disabil Rehabil 2000; 22: 170–180.
- 19. Tarvonen-Schroder S, Tenovuo O, Kaljonen A, Laimi K. Comparing disability between traumatic brain injury and spinal cord injury using the 12-item WHODAS 2.0 and the WHO minimal generic data set covering functioning and health. Clin Rehabil 2018 Jun 1 [Epub ahead of print].
- 20. WHO. International Classification of Functioning, Disability and Health. Geneva: World Health Organization; 2001.
- Cieza A, Oberhauser C, Bickenbach J, Chatterji S, Stucki G. Towards a minimal generic set of domains of functioning and health. BMC Public Health 2014; 14: 218.
- 22. Üstün TB KN, Chatterji S, Rehm J. Measuring Health and Disability: Manual for WHO Disability Assessment Schedule (WHODAS 2.0). Geneva: World Health Organization; 2010.
- Dijkers M. Quality of life after spinal cord injury: a meta analysis of the effects of disablement components. Spinal Cord 1997; 35: 829–840.
- 24. Huang WC, Wu TF, Liao HF, Liou TH, Kang YW. The relationship of participation restriction with quality of life in individuals with spinal cord injury. Phys Med Rehabil Int 2016; 3: id1099.
- 25. Middleton J, Tran Y, Craig A. Relationship between quality of life and self-efficacy in persons with spinal cord injuries. Arch Phys Med Rehabil 2007; 88: 1643–1648.
- Silva AG, Queiros A, Sa-Couto P, Rocha NP. Self-reported disability: association with lower extremity performance and other determinants in older adults attending primary care. Phys Ther 2015; 95: 1628–1637.
- Goldstein LB, Samsa GP, Matchar DB, Horner RD. Charlson Index comorbidity adjustment for ischemic stroke outcome studies. Stroke 2004; 35: 1941–1945.
- Kirshblum S, Waring W 3rd. Updates for the international standards for neurological classification of spinal cord injury. Phys Med Rehabil Clin N Am 2014; 25: 505–517.
- 29. Andrews G, Kemp A, Sunderland M, Von Korff M, Ustun T. Normative data for the 12-item WHO Disability Assessment Schedule 2.0. PloS One 2009; 4: e8343.
- Federici S, Bracalenti M, Meloni F, Luciano JV. World Health Organization disability assessment schedule 2.0: an international systematic review. Disabil Rehabil 2017; 39: 2347–2380.
- 31. Kuo CY, Liou TH, Chang KH, Chi WC, Escorpizo R, Yen CF, et al. Functioning and disability analysis of patients with traumatic brain injury and spinal cord injury by using the world health organization disability assessment schedule 2.0. Int J Environ Res Public Health 2015; 12: 4116–4127.
- Ahmed N, Quadir MM, Rahman MA, Alamgir H. Community integration and life satisfaction among individuals with spinal cord injury living in the community after receiving institutional care in Bangladesh. Disabil Rehabil 2018; 40: 1033–1040.
- 33. Kader M, Perera NKP, Sohrab Hossain M, Islam R. Sociodemographic and injury-related factors contributing to activity limitations and participation restrictions in people

Journal of Rehabilitation Medicine

46 S. Tarvonen-Schröder et al.

with spinal cord injury in Bangladesh. Spinal Cord 2018; 56: 239–246.

- 34. van der Zee CH, Post MW, Brinkhof MW, Wagenaar RC. Comparison of the Utrecht Scale for Evaluation of Rehabilitation-Participation with the ICF Measure of Participation and Activities Screener and the WHO Disability Assessment Schedule II in persons with spinal cord injury. Arch Phys Med Rehabil 2014; 95: 87–93.
- 35. Whiteneck G, Meade MA, Dijkers M, Tate DG, Bushnik T, Forchheimer MB. Environmental factors and their role in participation and life satisfaction after spinal cord injury. Arch Phys Med Rehabil 2004; 85: 1793–1803.
- Hossain MS, Rahman MA, Bowden JL, Quadir MM, Herbert RD, Harvey LA. Psychological and socioeconomic status, complications and quality of life in people with spinal cord

injuries after discharge from hospital in Bangladesh: a cohort study. Spinal Cord 2016; 54: 483-489.

- Schopp L CM, Hagglund K, Sherman A, Stout B, Gray D, Boninger M. Life activities among individuals with spinal cord injury living in the community: perceived choice and parceived barriers. Rehabil Psychol 2007; 52: 82–88.
- Derrett S, Beaver C, Sullivan MJ, Herbison GP, Acland R, Paul C. Traumatic and non-traumatic spinal cord impairment in New Zealand: incidence and characteristics of people admitted to spinal units. Inj Prev 2012; 18: 343–346.
- Saltychev M, Tarvonen-Schroder S, Barlund E, Laimi K. Differences between rehabilitation team, rehabilitants, and significant others in opinions on functioning of subacute stroke survivors: Turku ICF study. Int J Rehabil Res 2014; 37: 229–235.